Author's response to reviews

Title: Endoscopic Application of n-Butyl-2 Cyanoacrylate (Histoacryl) on Oesophagojejunal Anastomotic Leak: Report of a Case.

Authors:

Manousos- Georgios Pramateftakis (mpramateftakis@hotmail.com)
Georgios Vrakas (georgiosvrakas@gmail.com)
Ioannis Kanellos (ik@hol.gr)
Ioannis Mantzoros (mantzor-@otenet.gr)
Stamatis Angelopoulos (stamatis.angelopoulos@gmail.com)
Efthymios Eleftheriades (gvr156@hotmail.com)
Charalampos Lazarides (gatos69@gmail.com)

Version: 2 Date: 18 October 2010

Author's response to reviews:

Dear Sir/Madam,

Please find below the changes applied to our document. All changes are highlighted in the text:

1. The suggested reference has been added to the reference list.
2. The term “hazy” has been replaced by the term “turbid”.
3. The word “persisted” has been applied as suggested.
4. The timeline of the postoperative events has been made clearer in our text, by adding exact postoperative days.
5. We have improved the flow of written English, as seen on the highlighted sentences in the text.